Author's response to reviews

Title: HIV Testing and Care in Canadian Aboriginal Youth: A community based mixed methods study.

Authors:

Judy E Mill (judy.mill@ualberta.ca)
Randy C Jackson (randyj@caan.ca)
Catherine A Worthington (cworth@ucalgary.ca)
Chris P Archibald (chris_archibald@phac-aspc.gc.ca)
Tom Wong (tom_wong@phac-aspc.gc.ca)
Ted Myers (ted.myers@utoronto.ca)
Tracey Prentice (tpren087@uottawa.ca)
Susan Sommerfeldt (susan.sommerfeldt@nurs.ualberta.ca)

Version: 4 Date: 1 October 2008

Author's response to reviews:

Dear Editorial Production Team,

We have gone over the entire manuscript carefully and have changed minor typographical things. e.g., more accurate APA style guidelines i.e. double quotations to single quotations (vice versa) and where a numeric 6 became a "six". One word change on page 6 above the Methods title, we changed the word "quotations" to "quotes". On page 8, first sentence, we added the word " version 11.5" and the company and place in brackets ( SPSS Inc, Chicago, Il).

Yours Truly,

Judy Mill